Author’s response to reviews

Title: Doxapram alleviates low S p O2 induced by the combination of propofol and fentanyl during painless gastrointestinal endoscopy

Authors:

Xin Lian (tomytomytomy123@sohu.com)
Xin Lian (tomytomytomy123@sohu.com)
Haoxing Wang (wanghaoxing2005@126.com)
Chunxiao Hu (18626311198@163.com)
Zhiping Wang (zhpsqxt@163.com)
Shunmei Lu (lushunmei2008@163.com)
Jingjing Xu (jingjing6104@126.com)
Yiling Qian (ylqian-1987@163.com)
Jun Wang (wangjun19710930@126.com)

Version: 5 Date: 22 Sep 2019

Author’s response to reviews:

1. We have stated the study adheres to CONSORT in line 92.

2. We revised the authors’ contributions according to guidelines for authorship in line 284-287.

3. We added trail registration including the data at the end of Abstract in line 49-51.

4. We provided "ethics approval and consent to participate" in the Declaration section in line 271-272.

5. We uploaded our manuscript as a single, final, clean version according to comments.